

RESEARCH ARTICLE

Community Knowledge, Perceptions, and Practices Associated with Urogenital Schistosomiasis among School-Aged Children in Zanzibar, United Republic of Tanzania

Bobbie Person¹*, Said M. Ali², Faiza M. A'Kadir³, Jamal N. Ali³, Ulfat A. Mohammed², Khalfan A. Mohammed³, David Rollinson⁴, Stefanie Knopp^{4,5,6}

1 The Schistosomiasis Consortium for Operational Research and Evaluation, University of Georgia, Atlanta, Georgia, United States of America, 2 Public Health Laboratory—Ivo de Carneri, Chake Chake, Pemba, United Republic of Tanzania, 3 Zanzibar Neglected Tropical Diseases Control Program, Ministry of Health, Zanzibar Town, Unguja, United Republic of Tanzania, 4 Wolfson Wellcome Biomedical Laboratories, Department of Life Sciences, Natural History Museum, London, United Kingdom, 5 Swiss Tropical and Public Health Institute, Basel, Switzerland, 6 University of Basel, Basel, Switzerland

* bobbieperson@gmail.com

Abstract

Background

On the Zanzibar islands, United Republic of Tanzania, elimination of urogenital schistosomiasis is strived for in the coming years. This qualitative study aimed to better understand community knowledge, perceptions, and practices associated with schistosomiasis among school-aged children on Unguja and Pemba islands, in order to inform the development of behavior change interventions contributing to eliminate urogenital schistosomiasis.

Methodology

In 2011, we conducted 35 children's discussion groups, 41 in-depth interviews with parents and teachers, and 5 focus group discussions with community members in Zanzibar. Using a modified-grounded theory approach, we transcribed and coded the narrative data followed by thematic analysis of the emergent themes.

Principal Findings

Urogenital schistosomiasis is a common experience among children in Zanzibar and typically considered a boys' disease. Children engage in multiple high-risk behaviors for acquiring schistosomiasis because of poor knowledge on disease transmission, lack of understanding on severity of disease-associated consequences, and lack of alternative options for water related activities of daily living and recreational play. Local primary school teachers had little to no training about the disease and no teaching tools or materials for students.



Citation: Person B, Ali SM, A'Kadir FM, Ali JN, Mohammed UA, Mohammed KA, et al. (2016) Community Knowledge, Perceptions, and Practices Associated with Urogenital Schistosomiasis among School-Aged Children in Zanzibar, United Republic of Tanzania. PLoS Negl Trop Dis 10(7): e0004814. doi:10.1371/journal.pntd.0004814

Editor: Rachel L Pullan, London School of Hygiene and Tropical Medicine, UNITED KINGDOM

Received: December 7, 2015

Accepted: June 8, 2016

Published: July 11, 2016

Copyright: © 2016 Person et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Our study analyzes qualitative data and the participants did not consent to have their full transcripts made publicly available. The data policy exception related to privacy concerns pertains in this case. Excerpts of the transcripts relevant to the study are available on request from the corresponding author Bobbie Person at <u>bobbieperson@gmail.com</u>.

Funding: This study received financial support from the University of Georgia Research Foundation Inc., which is funded by the Bill & Melinda Gates

Foundation for these Schistosomiasis Consortium for Operational Research and Evaluation (SCORE) projects (prime award no. 50816, sub-award nos. RR374-053/4893206). SK is financially supported by sub-award no. RR374-053/4893196. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

Conclusions/Significance

Conducting activities in open natural freshwater contaminated by *S. haematobium* larvae compromises the health of school-aged children in Zanzibar. The perception of urogenital schistosomiasis as a minor illness rather than a serious threat to a child's well-being contributes to the spread of disease. Understanding community perceptions of disease along with the barriers and facilitators to risk reduction behaviors among children can inform health promotion activities, campaigns, and programs for the prevention, control, and elimination of urogenital schistosomiasis in Zanzibar.

Author Summary

On the Zanzibar islands, United Republic of Tanzania, elimination of urogenital schistosomiasis, a disease caused by infection with a blood fluke (Schistosoma haematobium), locally known as kichocho, is strived for in the coming years. This study used qualitative research methods of focus groups and in-depth interviews with adults, and group discussions with school-aged children to explore (i) knowledge and perceptions of kichocho transmission, (ii) specific behaviors among children in Zanzibar that put them at risk for acquiring infections with the kichocho parasite, (iii) symptoms of kichocho along with personal health-seeking behaviors and treatment strategies, and finally (iv) ideas for preventing kichocho in children. We found that there was little available formal education about disease transmission, which contributed to myths and misperceptions about routes of transmissions, causes, and treatment of kichocho. School-aged children regularly exposed themselves to contaminated natural, open freshwater bodies through recreational and domestic activities of daily life. Kichocho was often wrongly viewed as an infection with an intestinal worm of little significance, rarely associated with severe health consequences, and with little to no disease stigma. Local primary school teachers had little to no training about the disease and no teaching tools or materials for students. The findings add valuable insights into how current knowledge, perceptions, and practices impede optimal disease prevention and control and highlight the necessity for a community tailored behavioral intervention to interrupt transmission of urogenital schistosomiasis.

Introduction

Schistosomiasis is a debilitating disease that affects poor and deprived population groups, especially in rural Africa [1]. The global impact is enormous with more than 200 million people infected with blood flukes of the genus *Schistosoma* [2]. Urogenital schistosomiasis, caused by *S. haematobium*, can include acute illness such as blood in urine (hematuria) and anemia in children, while fibrosis of the bladder and ureter, and kidney damage can occur as infections persist [3,4]. Bladder cancer can be a complication in adults and female genital schistosomiasis may be associated with increased risk of human immunodeficiency virus (HIV) infection [5–9].

Over the past decades, programs to reduce the morbidity caused by schistosomiasis have been implemented in many endemic countries and the number of people who received treatment with praziquantel has increased annually [10]. In order to eliminate schistosomiasis as a public health problem and to interrupt transmission in areas where morbidity control has been achieved, the World Health Organization (WHO) and other institutions and stakeholders are advocating the intensified use of integrated schistosomiasis control approaches [7,11-14]. Health education and interventions based on social and behavioral science triggering behavior change, in addition to regular preventive chemotherapy with praziquantel, are likely to become a key component of future elimination efforts [12,13,15].

Behavior change in humans requires close interaction with the at-risk population [16]. To change risk behaviors, enhance the interruption of disease transmission, and finally eliminate schistosomiasis particularly in children, innovative, community-tailored approaches are needed. Understanding the community is critical to creating effective behavioral interventions promoting the adoption of protective behaviors and reducing risk behaviors [16,17]. Community participatory processes are fundamental to understanding the community's current knowledge, perceptions, attitudes, and behaviors as well as motivators and barriers to behavior change [18]. Community participation can also create ownership of public health initiatives, which is often viewed as fundamental for the success of population-based health outcomes [16,18]. People are experts about the communities they live in and have many different ways of knowing and gathering information.

On the Zanzibar islands, United Republic of Tanzania, elimination of urogenital schistosomiasis is strived for in the coming years. In the frame of a randomized operational research trial implemented on Unguja and Pemba islands from 2012 till 2017, the impact of behavior change interventions in addition to biannual praziquantel treatment on the prevalence and intensity of *S. haematobium* infections is assessed [<u>19–21</u>]. Here we present results of the qualitative formative research that was implemented in 2011 to inform, along with future participatory community co-design workshops, the development of a community-tailored behavioral change intervention that might help to eliminate urogenital schistosomiasis in Zanzibar [<u>19</u>].

Methods

Ethics Statement

In 2011, the National Centre for Emerging Zoonotic Diseases (NCEZID) of the Centers for Disease Control and Prevention (CDC) received and approved the qualitative formative research protocol for Zanzibar (NCEZID Tracking Number: 103111BP) to go forward to the CDC Human Research Protection Office (HRPO) and Institutional Review Board (IRB) for review. The HRPO and IRB determined that the project activities were exempt under regulation 45 CFR 46.101(b)(2) and issued a written waiver. The full study protocol of the "Study and implementation of schistosomiasis elimination in Zanzibar (Unguja and Pemba islands) using an integrated approach" received additional ethical approval from the Zanzibar Medical Research Ethics Committee in Zanzibar, United republic of Tanzania (reference no. ZAMREC 0003/Sept/011), the "Ethikkommission beider Basel" (EKBB) in Basel, Switzerland (reference no. 236/11) and the IRB of the University of Georgia, in Athens, Georgia, United States of America (project no. 2012-10138-0). The study is registered at the International Standard Randomized Controlled Trial Number Register (ISRCTN48837681). The data collection was conducted with support from the CDC in Atlanta, Georgia, United States of America. The CDC HRPO and IRB approved the informed consent process conducted with all participants, who took part in student group discussions and interviews. Due to a limited ability of participants to read and write the informed consent was available in both English and Kiswahili, the local language, and read aloud by trained bilingual research staff. Participants provided a verbal consent, with the consent acknowledged with the signature on the informed consent document of a witness present at the time [22,23]. Research staff reviewed the consent procedure and all consent forms to ensure compliance with the process. In case of children below the age

of 18 years, their parents or legal guardians provided written informed consent for their participation.

Study Area and Population

This qualitative inquiry was conducted including school-aged children, parents, teachers, and community leaders from seven small administrative areas, called shehias, on the islands of Unguja and Pemba from July till September 2011. The islands of Unguja and Pemba have an estimated combined population of around 1.3 million people and the main industries are spices, raffia, and tourism [24,25]. More than 99 percent of Zanzibar's population is Muslim. Urogenital schistosomiasis constituted a considerable public health problem on both islands in the past century [26–28], but regular treatment of the at risk population with praziquantel reduced *S. haematobium* prevalences and intensities [29–31]. In 2012, the baseline survey of the "Study and implementation of schistosomiasis elimination in Zanzibar (Unguja and Pemba islands) using an integrated approach" revealed an overall prevalence of 4.3% and 8.9% in schoolchildren from Unguja and Pemba, respectively [20].

Research Team

The field team consisted of a senior social scientist from the CDC and seven Kiswahili and English speaking research assistants from the Ministry of Health, Department of Neglected Tropical Diseases, and the Ministry of Education, Department of Health Education, in Unguja, and three research assistants from the Public Health Laboratory—Ivo de Carneri in Pemba. The local team had little or no previous experience with the application of qualitative research methods. Research assistants were trained in research ethics and qualitative data collection methods by the senior social scientist and served as the primary data collectors and logistic coordinators setting up focus groups and interviews within the communities.

Recruitment of Participants

We used purposive sampling to recruit a homogeneous study sample of school-aged children, who might engage in risky behaviors. Such risk behaviors include swimming, fishing, bathing, washing clothes, or performing other domestic chores in ponds, lakes, streams, and rivers that are potentially contaminated with *S. haematobium* larvae. For this initial informative research study it was decided that the easiest way to reach school-aged children was through government supported public primary schools. Schools in the selected shehias were identified with the assistance of the staff from the Ministry of Health and from the Ministry of Education. Students in grades Standard 1 to Standard 7 were recruited through the headmaster of each school. Individual teachers and parents were also recruited through public primary schools and community leaders were recruited through local social networks.

Data Collection

The qualitative inquiry was conducted through 35 children's group discussions (GD), 5 focus group discussions (FGD) with community leaders, and 41 in-depth interviews (ID) with teachers and parents. The children's GD included 6–8 children of the same sex, facilitated by a Kiswahili speaking research assistant using a simple topic guide to lead the discussion (S1 Topic Guide). Children were provided paper and crayons and were first asked to draw anything they wanted to draw, followed by a discussion of their picture. Then they were asked to, *"Please draw me a picture about everything you know about kichocho (the local term for schistosomiasis)."* These drawings encouraged more robust discussions [32]. Students drew pictures and

then described their drawings about the disease, risk behaviors, and prevention ideas. A note taker managed an audio recorder and took written notes in support of the group facilitator. FGD with community leaders followed a similar format without the drawings. Individual IDIs were conducted with parents and teachers using the audio recorders without the note takers. Examples of qualitative open-ended questions used with the adults are shown in Fig 1. The study was conducted in five shehias on Unguja (Chaani, Dole, Kilombero, Mwera, and Uzini) and in two shehias on Pemba (Chambani and Kizimbani). The shehias were selected among the 15 behavioral study shehias on each island [19], based on their location on the island and previous knowledge about urogenital schistosomiasis in the area [29,31,33,34]. Data were collected until saturation was obtained [35,36].

All data collection tools including project overview, informed consents, and topic guides were translated from English into Kiswahili, pretested, and modified to adapt to local linguistic and cultural nuances by the research team [37]. Topic guides explored i) knowledge and perceptions of schistosomiasis transmission, ii) specific behaviors among children in Zanzibar that put them at risk for acquiring *S. haematobium* infections, iii) symptoms of schistosomiasis along with personal health-seeking behaviors and treatment strategies, along with iv) ideas to prevent schistosomiasis in children. Additionally, we explored other cultural factors, gender-roles, influential communication channels, and decision-making processes [37]. Additional probes allowed for deeper exploration of the topics that emerged, supporting additional areas of interest. Following

. Please	tell me all you know about kichocho.
	re some possible probes for discussion: (If they don't talk about these issues, probe with the questions below.)
Yo	I have told me that you get kichocho from the river. How does one get kichocho from the river? What is in the river or pond that causes kichocho
	ere did you learn this? What have you <u>heard</u> about kichocho? Where did you <u>hear</u> it?
	o can get kichocho? Boys/girls? Men/women?
	at are the symptoms of kichocho for boys and men? What are the symptoms of kichocho for girls and women?
	there other conditions that can cause blood in the urine?
WI	at types of health messages have you heard about kichocho? Where did you hear them? Who sponsored them?
2. Please	tell me about you or anyone in your household who has had kichocho.
Here a	e some possible probes for discussion: (If they don't talk about these issues, probe with the questions below.)
WI	o had kichocho? Had they had kichocho before? How did you or your family know it was kichocho? What were the symptoms?
	at did you do when you (or someone else) started feeling pain and/or urinating blood?
	ase tell me about any home remedies that can be used for kichocho. Please tell me about any traditional healers that treat kichocho.
Ho	v have you tried to prevent getting kichocho again? What do you differently now? How much does it cost to treat kichocho?
8. Please	tell me about what behaviors or habits that children (boys/girls) do that lead them to get kichocho.
Here a	re some possible probes for discussion: (If they don't talk about these issues, probe with the questions below.)
Ho	v often are they in the river or pond? What do they do at the river? Tell me more about how they wash their clothes.
WI	ere else do they play? What do they do? Where do they urinate? Do you think they ever urinate in the river or pond?
Ho	v can we get them to not urinate in the water?
4. Please	tell me what you think could be done in order to prevent children from getting kichocho?
How o	an we change those behaviors of urinating in the water, playing and bathing in infected water?
Here a	re some possible probes for discussion: (If they don't talk about these issues, probe with the questions below.)
	at can we give children to do instead of playing in the river? What games do they like? What activities do they like to do?
5. Teach	rs Only: Please describe to me any education curriculums or materials you have used to teach children about kichocho.
	re some possible probes for discussion: (If they don't talk about these issues, probe with the questions below.)
	at did/do you like about these materials? What did you dislike?
	at would you like to have for your classroom to educate children about kichocho?
	at groups or clubs in your school would be good to teach about kichocho?
	at groups of clubs in your school would be good to teach about Nichocho?

Fig 1. Sample of qualitative probes for adults participating in a qualitative study about schistosomiasis in Zanzibar.

doi:10.1371/journal.pntd.0004814.g001

group discussions and interviews, participants received a small thank you gift for their time and participation. Group discussions conducted in schools and FGD conducted in community settings were approximately 1.5 hours in length, IDIs took approximately 45–60 minutes. Data were collected using audio recorders (Olympus 70, Olympus Corporation, Tokyo, Japan). Additional field notes were handwritten during the interviews and reviewed during debriefing sessions to verify accuracy of the interview session [<u>38</u>]. We did not collect personal identifiers.

Data Management and Analysis

Due to the paucity of behavioral information on urogenital schistosomiasis in Zanzibar we chose a qualitative approach to better understand cultural practices associated with activities of daily living linked with contact with local natural open freshwater bodies. In this study, we examined barriers to schistosomiasis prevention and control related to urination practices of children along with recommendations for improving such practices and reducing disease threats. We used a modified grounded theory approach with an emergent qualitative thematic analysis allowing the hypothesis to be generated from the data [39].

Narrative data were transcribed into English with review following translation to ensure accurate translation and local meanings. Transcripts were entered into Atlas-ti (ATLAS.ti Scientific Software Development GmbH, Berlin, Germany) as a Word document (Microsoft Corporation, Redmond, WA, USA) to facilitate text searching, data coding and analysis. Data analysis began with the first discussions and interviews allowing for emerging, unexpected, and/or inconsistent issues to be explored in subsequent discussions and interviews [36,39]. The coding structure evolved inductively with the codes from the narrative data of earlier interviews informing subsequent coding of the following interviews supplemented with field notes from the interviewer and note taker [32,38-40]. Due to time constraints and ongoing data collection tasks, the primary author, a social scientist experienced in qualitative research, was the primary data coder with verification of interpretive codes by the research assistants. Open, axial, and selective coding was used to analyze the GD, FGD, and IDI narratives [36,38-41]. A coding frame was developed through open coding, a word-by-word analysis used to identify, name, and categorize explanations and descriptions of the day-to-day reality of participants as related to schistosomiasis. Consensus on the coding frame was obtained through discussions with the local qualitative research assistants, who were from Zanzibar and conducted the original interviews. Axial coding, the process of relating codes to each other, via a combination of inductive and deductive thinking, was used for analysis of specific emergent themes, across themes, and for the relationships between themes [40,41]. Over the course of data collection, emergent themes became redundant, suggesting that all major themes had been identified and saturation reached [42]. An analysis matrix served as a framework for the resulting findings. Narrative excerpts from an analysis framework matrix are shown in Table 1.

The trustworthiness of our data was derived from standardization of methods and documentation for auditability, triangulation of the data, and verification of data findings with local staff members [43]. A standardized implementation document guided the training and implementation of the qualitative methodology with all procedures, topic guides, informed consents, timelines, interview schedules, data collection strategies, data management, and analysis strategies written out [38,43]. Triangulation of data was derived through the multiple data collection methods (GD, FGD, and IDI), multiple perspectives (younger and older girls and boys as well as adult women and men), and multiple venues (school-based, private home-based, and public venues). Findings were verified amongst local staff of the Ministry of Health in Unguja and the Public Health Laboratory Ivo-de-Carneri in Pemba by corroborating results with similar findings across other settings [44,45].

Qualitative method	Quotations from individuals										
GD Older Boys; GD Young Boys	We were told that kichocho is transmitted in the river. There are <u>bacteria</u> that cause kichocho. I think we have to educate our friends that kichocho is affecting the human body. K2; ** explore bacteria	There are blood sucking worms, crabs and <u>snails</u> in the river that can all cause kichocho. M1	Kichocho is a disease that is transmitted if someone has gone to toilet and another person who doesn't have kichocho steps there in bare feet. K3; ** common myth	A person can get kichocho by <u>eating chilies</u> . Eating chilies causes kichocho. U2; **many people have mentioned this							
GD Older Girls; GD Young Girls	Sometimes one may take water for brushing the teeth and when you gargle with the water there are organisms that will get into you. That's how people get kichocho. U1	Kichocho is caused by the "bakari kichwa" (young frog). There are also small fish that bite and cause kichocho. C2	Girls don't get kichocho by going in the river. They are very few. The majority of girls get kichocho by <u>eating chilies</u> . M2; ** many people have mentioned this	A <u>snail bite</u> s you and you get sick with kichocho. K2							
Community Member FGD	There are organisms in the river that may <u>enter in the mouth</u> from the way the children swim. They cause kichocho. U-FGD	Myself, I have never heard of it until I moved here. They told me it is transmitted by <u>snails</u> in the river. But I really know nothing. C-FGD	Kichocho is a disease that affects children and even adults when they go swimming in dirty water with small fish which cause kichocho. K-FGD	I don't know if the exact cause of kichocho is the river or not. Maybe it is the <u>water we drink.</u> M-FGD							
Teacher IDIs	You will get kichocho from organisms which live in <u>snails.</u> They penetrate through a person's skin especially when you enter the water. D4	Most children get [intestinal] worms. This kichocho worm. It is different. I suffered kichocho when I was a boy as well. It is that way. C5	You know in some rivers and ponds there are <u>snails</u> that cause kichocho by producing small organisms that get into the human body causing the disease. P1	A person who baths in that river he will take on the kichocho disease. There is this <u>bacterium</u> in the river that causes kichocho. U3; ** explore bacteria							
Parents IDIs	Kichocho is a disease which is caused by dirtiness from the river. When people go to the river for their activities like washing clothes they make it dirty. C1	the transmission can be between husband and wife, you know. K4	Kichocho can come from the soil or water or when the water is stagnant. These things they stick to the soles of your feet. M2	When you <u>drink the water</u> from the river you will swallow the tiny organisms and get kichocho. M1							

Table 1. Narrative quotations from analysis matrix #2 to explore people's perception of what causes schistosomiasis (kichocho).

FGD: Focus group discussion; GD: group discussion; IDI: In-depth interview

doi:10.1371/journal.pntd.0004814.t001

Results

Study Characteristics

As shown in <u>Table 2</u>, narrative data were collected from 27 children's GD, 5 FGD with community leaders, and IDI with 21 teachers and 16 parents on Unguja Island. We also conducted 8 children's GD and 4 IDI with teachers on Pemba Island to verify similarities among children

Table 2. Characteristics of groups involved in a qualitative study about urogenital schistosomiasis in Zanzibar.

	Children			Parents		Teachers			Community leaders			
	Female	Male	Median age (range)	Female	Male	Median age (range)	Female	Male	Median age (range)	Female	Male	Median age (range)
Unguja												
Group discussions (n = 27)	78	81	11 (6–17)				19	2	37 (26–56)			
Focus group discussions (n = 5)										14	16	46 (23–72)
In-depth interviews				8	8	40 (24–72)						
Pemba												
Group discussions (n = 8)	25	21	11 (8–15)									
In-depth interviews							2	2	35 (26–52)			

doi:10.1371/journal.pntd.0004814.t002

on the two islands. Group discussions included boys and girls from both islands, who were 6–17 years old and attended grades of Standard 1–7. FGD included 16 male and 14 female community leaders, aged 23–72 years. They were teachers, farmers, leaders of women's groups, religious leaders, school coaches, religious schoolteachers, petty traders, and small business owners, as well as housewives and traditional leaders. In-depths interviews were conducted with 2 male and 19 female teachers from Unguja and 2 female and 2 male teachers from Pemba. Teachers were 26 to 56 years of age. Eight fathers and eight mothers from Unguja were interviewed. Parents had a median age of 41 years with a range of 24 to 72 years of age.

Emergent Themes

Perception of Schistosomiasis transmission. Children and adults often identified "the river," generic organisms, bacteria, small frogs, blood-sucking worms, crabs, dirt, other people's urine, with few people reporting snails as part of the schistosomiasis transmission cycle. Young children often reported that leeches and small frogs caused kichocho. Older boys, many with previous experience of kichocho symptoms, reported receiving information about disease transmission from doctors, friends, or someone at school. An older male student said, "We got this disease and when we went to hospital the doctor educated us about this so we can prevent it in ourselves." (GD_K1). Despite this, most boys had a poor understanding about disease transmission. Children described several widespread myths about kichocho transmission. An older student said, "Kichocho is a disease that is transmitted if someone has gone to toilet and another person who doesn't have kichocho steps there bare feet." (GD_K3). Another older girl also reported, "Sometimes one may take water for brushing the teeth and when you gargle with the water there are organisms that will get into you... that's how people get kichocho." (GD_U1). Many teachers reported that they had limited knowledge of kichocho because there was only a small section in local science books. Teachers lacked a standardized, detailed syllabus to teach children. Only a couple of teachers had accurate knowledge about disease transmission, one reported, "... you will get kichocho from organisms which live in snails. They penetrate through a person's skin especially when you enter the water." (Int_D4). Parents knew the least about the transmission and complications of disease. A couple of parents told us that kichocho could be transmitted man to woman during sexual intercourse. A parent said, "... the transmission can be between husband and wife, you know." (Int_K4). Overall, people perceived the infection as being insignificant, associated with worms that everyone has at one time or another. A male teacher told us, "Most children get [intestinal] worms. This kichocho worm... I suffered kichocho when I was a boy as well. It is that way." (Int._C5)

Risky practices of children. Kichocho was characterized by children and adults as a boy's rather than a girl's disease because boys spend more time in contaminated waters. Parents, teachers, and community leaders identified playing, bathing, fishing, and other water activities as risky behaviors of boys. A teacher reported, "Small boys go there [to the river] with their parent. . . and others follow their friends in groups because they are told by their parent to go and bath at the river so they should not spoil water at home. Water is very little and difficult to get. . ..boys will bath for two hours or more in the river." (Int_K3). A younger boy reported, "When we go to the river it is fun! On Sunday there is no tap water so if you want to wash school uniforms you have to go there. When you go there you will pretend to put the clothes in the water but you are there to have fun in the water." (GD_M2).

Most adults and some children associated the act of urinating in the river as a behavior that puts others at risk for disease. They described boys as urinating in the river more frequently than girls. Boys also told us that they urinated in the river, ponds, and fields. An older boy reported, "*Boys don't like to urinate in the house. They urinate everywhere... even in the bushes.*

Men aren't shy. Men will urinate anywhere. Especially the young ones, they will urinate in the water, whereas an elder hides when finding a place to urinate." (GD_K2). Parents in every community acknowledged that boys urinate everywhere. A parent reported, "When a boy is at home he uses the toilet, but if he is somewhere else he will urinate anywhere. I have been through all that. When I was a child I urinated in the river and everywhere." (Int_D3).

Most adults reported that girls engage in fewer risky behaviors than boys. Teachers and parents told us that girls play in the river like boys until the age of 6 or 7 at which time they are separated from activities with boys, in line with their Muslim culture. Most risky activities were associated with domestic chores such as washing clothes and dishes as well as carrying water. An older girl said, ". . . *girls go swimming, wash clothes and dishes [at the river] and take water home for domestic chores.*" (*GD_C2*). But another older girl told us that girls are not at risk for kichocho, "*They [girls] don't void blood. They don't get kichocho.*" (*GD_K9*). Most adults, across all communities, described girls as using a toilet when urinating at home and urinating in the field when the need to urinate is urgent. A teacher told us, "*Girls urinate at their home and fields, very rarely in the rivers.*" (*Int_C2*). Yet, some adults believed that girls do urinate in the river. A community leader told us that girls do urinate in the river, "*If girls are in the rive they will urinate there.*" (*Int_M3*). A young girl also told us, "*Girls are different from boys. They may be a little shy, but they do urinate, but not like boys. Girls will hide in the water but boys will pour it openly.*" (*GD_M1*).

Symptoms of kichocho. Most children and adults reported that kichocho symptoms were typically bloody urine with pain. An older boy reported, "*The urine will not come out and finally you will see blood coming out from the private part. You will get severe pain and at last there is a pus discharge and blood at the end when you finish." (GD_K1). A parent with sons said, "For kichocho there is severe pain at the kidney site. Sometimes urine may be leaking, itching, then bleeding. Signs are the same to boys and men but when the adult is sick he will go to the hospital himself. A boy will not say until the problem becomes chronic." (Int_M4).*

Some adults and children reported uncertainty as to whether symptoms were different between adults and children and boys and girls. A teacher said, "I am not sure if they [symptoms] are different between children and adults. I think they are all the same. I am not sure. I don't know." (Int_C4). A community member also reported, "I think you feel pain at the urinary bladder. Also [there will be] blood in the urine because the disease is the same. The symptoms will differ according to the different nature of the male and female. Therefore, I don't know which symptoms are for male and which symptoms are for female." (FGD_D1). Adults and children failed to identify kichocho symptoms for girls and women. One parent reported, "To them [girls] I think it would be very difficult to know [if she has kichocho]. For us, it is easy because we see it. However, for girls how would they know? Maybe when they wash their underwear and see blood they will know." (Int_U3).

Community members we interviewed also provided a variety of other explanations for blood in the urine that could lead to delayed health-seeking behaviors. Many adults reported that gonorrhea, syphilis, and other sexually transmitted diseases, cancer, childbirth, and menstruation could cause blood in the urine. One man said, *"I think according to what we have heard, even a sexual transmitted disease like syphilis may cause blood in the urine." (Int_Z1)*. In addition, parents, community members, and teachers reported that heavy exercise and exertion due to hard work could cause blood and pus in the urine. A teacher said, *"Sometimes strenuous exercises can cause blood in the urine. Don't be surprised when a football player tells you, I have blood in the urine. I had this experience once." (Int_MK1). A couple of parents and community members attributed blood in the urine to witchcraft: <i>"You can bleed if someone has done bad witch to you." (Int_MW2)*. Several community members and parents and even a teacher attributed blood in the urine to eating local chilies.

Health-seeking behaviors and treatment. Many teachers, parents, and community members reported that while some people self-treat, most seek care at a hospital or clinic. Children were often reluctant to tell their parents they had kichocho. Costs of treatment varied, and even when free, people reported putting off seeking care because of transportation costs. Parents reported that they took children to local clinics only to find no available drugs for treatment. A referral to another facility, incurring more transportation costs, was a barrier to seeking treatment. A few parents, community members, and teachers described home remedies and traditional healers as first line care before going to seek care at a hospital for treating kichocho. Home remedies included preparing teas from the root of a plant or drinking copious amounts of water. A teacher reported, "My relative used hospital treatment but there are some kind of roots which are used by people....there are various roots for many diseases and kichocho can be treated by roots as well. There is one which is called "mkaanga uchawi" (frying a witch) or "mchafufu" some people call it a "tetracycline tree" because its fruits has two colors . . . red and black.they boil the root and mix it with some herbs and drink the soup. . . it can be taken three times a day . . . it depends." (Int_D4). A parent told us, "I had painful urine and blood. I had signs like people were saying. I went to three traditional healers but it was useless...until I went to hospital and got better." (Int_K2).

Mass drug administration in schools was often reported as the primary way children get treatment for kichocho: "Children get drugs for free when health workers from Kichocho Program come to school." Parents described various experiences when seeking treatment for their children with kichocho through other avenues. A community leader told us, "One has to go to hospital because sometimes local treatment does not help. You can be cured and get sick again so is better to be treated in hospital." (FGD_D1). Another parent said, "There was a time the drugs were free. . .now because of new technology one can go to private hospitals which treat Kichocho. . . you pay." (Int_M6). There were many ideas about the costs of treating kichocho. Some adults and children reported that treatment was free. A parent said, "We got treatment for free. . .we did not pay. It was good." (Int_M7). But many others reported various treatment-associated costs. A teacher reported, "The time I went [to the hospital] I got medicine freely. But transport cost from home to the center: going and returning." (Int_Z2). A student reported, "My friend was sick and we took him to the hospital where they gave him one tablet and he was asked to pay 400 shillings. . .one tablet." (GD_K1). A teacher also told us, "We buy [drugs for kichocho] in town at a private pharmacy. I remember a tablet cost about five hundred shillings." (Int_C3).

Few adults knew where to obtain free treatment for kichocho. A teacher reported, "I have heard that there is a special hospital for kichocho at Mianzini. I am not sure about the availability of drugs or if you have to buy or not." (Int_D2). Parents also expressed concern over having to pay transport even if treatment was free: "There are direct or indirect costs. An example of direct [cost] is buying tablets in private pharmacies or hospitals. An indirect [cost] is getting the tablets for free but paying bus fee to town." (Int_Z1).

Children's role models. Many adults identified parents, Madrassa (Koran school) teachers, and schoolteachers as role models for younger and older children. A parent said, "*Most of the time they [children] are with us. When you do something they follow. If it is good or bad...if you use abusive words they will do also. Girls follow the habit of their mothers and boys follow the habits of their fathers." (Int_D4). Some parents reported that they were the most important role models for their children but that it was very challenging and they needed help to change their children's behavior. "We have been trying very hard without success. Maybe you can support us...we cannot make it happen," said a parent. (Int_C2).*

Many teachers reported that the parent is a critical role model for their child. Teachers also told us that support is needed to teach parents how to prevent their children from getting kichocho. A teacher told us, *"First it is the parents that are important. They need to get*

knowledge. A child should be educated about rivers at home so he can change behavior. We can also use the Sheha [community leader] committee members. They can fight against kichocho because they are respected by the community." (Int_M8). Many other adults and also children acknowledged the role of a child's peers in influencing their behaviors. An older boy from Kilombero shehia reported, "The young ones [boys] they follow us. If they see us go to the river they will follow...even for fishing. So after a time he will get used to that and continue going to the river. If you say to him to stop going he will say...but you go. Why are you stopping me to go?" (GD_K3). A parent told us, "A child will easily follow the behavior of another child. When he sees him doing something he will do the same." (Int_M4). Additionally, it was acknowledged by most children that older boys are often the most influential role models, both positively and negatively, on the behaviors of younger boys.

Stopping urination in freshwater sources. Most adults and some children recognized the difficulty of extinguishing the behavior of urinating in the ponds and streams. It was seen as a private behavior and often associated with urgent need. A young student, when asked why he plays and sometimes urinates in the river when he knows not to reported, "I am driven because of "ubilisi" (the devil)." (GD M2). Most adults interviewed suggested fear and punishment to prevent children from urinating in the river even though they themselves had been forbidden to go in the river and did so anyway when young. A parent told us, "It is important to frighten them [children]. If you urinate in the river you will create and get diseases. If you urinate in the bush the devils will see you and you will be sick. We have to educate them to use toilet all the time even if they are outside." (Int_M2). A community member reported, "I think it's simple to talk about but it's difficult to change [behavior] because this boy in the bush, no one can see him [urinate]." (FGD_D1). A teacher reported that a comprehensive educational effort was needed to change behavior, "I think children need more health education. There should be street banners, more books on Kichocho in schools, and community mobilization like in malaria campaigns." (Int_C2). Another teacher suggested going beyond lectures and encouraging more engaging activities for children, "We can make small groups and teach them to prepare drama and role play to educate them about the effect of playing in the river. This would educate them not to urinate and bath in the river and ponds. for children, the drama and role plays retain *in their mind rather than words....and this will help to change their behavior.*" (Int_D3).

Children and adults across all groups interviewed suggested several structural interventions associated with changing the environment in order to facilitate behavior change in children. One teacher suggested, "We should build toilets. The community should be educated about toilet building especially in public areas like schools, so children can learn toilet-using behavior. But we have to have toilets." (Int_D4). A secondary student said, "People can get together in a meeting and decide to dig a latrine and a well so we can get water which mixes with soap easily and use it to wash our clothes. They [children] won't go to the river." (GD_K2). Another secondary student suggested creating special policies, "In our communities we do not use toilets. You should have a policy that everyone has to build a toilet in their home. It would help avoid using the water in the river." (GD_C2).

Preventing kichocho in children. Community leaders, teachers, as well as students often described the need for community members to work together against kichocho. All groups of adults as well as students identified two major channels for reaching children not attending school. The first was reaching out and educating all parents in the community and the second was through Madrassa schools. One older student reported, *"This problem [kichocho] is affect-ing us so much that the Ministry of Health has to prepare a syllabus for us in schools." (GD_C2).* Another student suggested, *"I think we have to be friends with them [children] not to scold them. We should find the technique of making them to forget the river and to remind them to go to school and Madrassa. That would be better." (GD_K3).*

Teachers provided a variety of ideas for preventing kichocho. One teacher said, "*The first thing teachers need is training to provide knowledge to the children about everything about kichocho.*" (*Int_W2*). Another teacher said, "*We need to create a syllabus concerning health and kichocho.*" (*Int_W1*). Teachers also expressed a need for children to have alternative options for spending their spare time and playing. A teacher told us, "*We should organize interesting sports like football for students. Toilets and latrines should be built and children should be punished when they play in the river." (<i>Int_MK1*). Parents also told us that children needed play options. One parent said, "*I think we should build playgrounds near the river so children will be attracted to play in the grounds instead of the river.*" (*Int_C4*). A number of community leaders suggested improving water sources and increasing access to clean water for their community. A community leader told us, "*We can prevent kichocho by building as many wells as possible for clean water.*" (*FGD_C1*).

Discussion

Qualitative research, alone or in mixed methods, has been used to better understand the experiences of people affected or at risk for numerous neglected tropical diseases such chagas disease, filariasis, and schistosomiasis [46-48]. Results of our informative in-depth, qualitative investigation of schistosomiasis among school-aged children suggested that despite previous initiatives related to urogenital schistosomiasis control and prevention in Zanzibar [29,44,49], people's knowledge about disease symptoms, transmission, and prevention were poor. Our findings identified several barriers to optimal disease prevention and control.

First, we observed that school-aged children regularly exposed themselves to contaminated natural, open freshwater bodies through recreational and domestic activities of daily living with little knowledge about routes of schistosomiasis transmission, which is in line with findings from previous studies in Zanzibar, Tanzania, Zimbabwe, and Western Kenya [50–52].

Second, S. haematobium infection was often viewed as an infection with an intestinal worm of little significance, not typically associated with severe health consequences, and little to no disease stigma. This is in contrast to reports from previous research in Nigeria, where individuals with schistosomiasis disease were stigmatized by others [53]. The Health Belief Model posits that perceived seriousness along with perceived susceptibility, perceived benefits, and perceived barriers are critical constructs used to explain and influence changes in health behaviors [54,55]. It also specifies that if individuals perceive a negative health outcome to be severe and perceive themselves to be susceptible to those negative outcomes, they are more likely to adopt positive protective behaviors [17,54,55]. Drawing upon the constructs of this behavioral theory supports shifting the context of schistosomiasis to that of a blood fluke, with serious health consequences such as bladder cancer and infertility, rather than the current perception of a less severe "worm." Elaborating on the perceived seriousness of the infection, whether through medical information or increased awareness of the serious effects of the disease on a person's life, is critical to address in a behavioral intervention [17,54,56]. There is evidence that theorybased, behavioral interventions can increase effectiveness among a variety of public health issues [57-60]. Synthesis of behavioral intervention research and non-regulatory interventions most often advocates the application of behavioral theory as an integral step in intervention design and evaluation [55,61].

Third, many people described abdominal pain, blood in the urine (hematuria), pain or burning during urination (dysuria), and commonly genital itching as symptoms of infection. However, as observed in studies conducted elsewhere in sub-Saharan Africa [62], these symptoms were also perceived as sexually transmitted infections that indeed may appear similar to symptoms of urogenital schistosomiasis. A person with a sexually transmitted infection may be reluctant to seek treatment due to shame and stigma [53,62]. Therefore, correcting the misperception that schistosomiasis is a sexually transmitted disease, while at the same time supporting the need to seek treatment for any and all similar symptoms, could be an important component of a schistosomiasis educational campaign to improve treatment seeking.

Fourth, first line treatment for a few people in Zanzibar, similar to mainland Tanzania [62], was often home remedies and occasional use of locally available herbalists followed by more conventional treatments when those earlier ones had failed. Lack of decentralized, locally available drugs and cost of transportation were also identified as barriers to seeking more conventional drug treatments. The decentralization of drug treatment to the local level as well as increasing knowledge about free drug treatment accessible through mass drug administration campaigns could improve treatment seeking among infected individuals. Further research into understanding any underlying barriers to treatment seeking behaviors should be explored [63].

Fifth, little available formal education about disease transmission contributed to myths and misperceptions about routes of transmissions, causes, and severity of disease, treatment, and ultimately prevention of disease. Schoolteachers and Koran school (Madrassa) teachers, viewed as influential people in children's lives lacked formal scientific training, teaching materials, and other resources to be able to educate students about schistosomiasis. Teachers reported a need for a teacher's training with a standardized, detailed syllabus to teach children about schistosomiasis during school sessions. Trainings could be set up similar to the Lushoto Enhanced Health Education Project that introduced interactive teaching methods into mainland Tanzanian study schools and demonstrated a feasible and effective intervention capable of changing schistosomiasis knowledge and health seeking behaviors among children [64]. The inclusion of religious teachers as change agents could maximize exposure of a schistosomiasis educational program to a broader community because they often engage children who may not attend government schools. Trained school and religious teachers could instill a perception of perceived seriousness of disease as well as perceived susceptibility of disease among children engaging in risky behaviors. Teachers could also identify and address the barriers to change and promote perceived benefits of reducing risky behavior to children.

Educating through schools could encourage students to act as change agents through peer education, role modeling, and shifting social norms of acceptable behavior [65,66]. Peer education, defined as "the teaching or sharing of health information, values and behaviors by members of similar age or status," is widely used in the field of health promotion and education recently, such as the prevention of HIV/acquired immune deficiency syndrome (AIDS), smoking, and alcohol and drug use [67–71]. Peer education is focused on sharing information and experiences along with trust between the people in the similar context and learning from each other. Peer education, has been noted as a feasible method for transferring schistosomiasis knowledge from students to parents [65,66].

Sixth, most adults, and some children recognized the difficulty of extinguishing the behavior of urinating in the ponds and streams. It was seen as a private behavior and often associated with urgent need. Children and adults described educational, behavioral, and structural interventions to prevent kichocho in children. Community members often described the need for the community to work together to prevent kichocho in children suggesting the importance of a participatory approach to intervention development and implementation.

Previous research has noted that top down approaches to community interventions have been perceived by some community members as not in their best interest or being a poor fit for the socio-cultural context within the community [72-74]. The lack of attention to an individual's social, cultural, religious, environmental, and physical context often results in a poor understanding of why an intervention is valuable and ultimately in an inadequate adoption of the desired positive behaviors and practices by community members [72-74]. This may explain why despite years of community administered preventive chemotherapy, the perception of schistosomiasis in Zanzibar was that of a commonplace, minor illness, rather than a serious threat to a child's wellbeing. Administering preventive chemotherapy without addressing the local circumstances of community members with tailored communication and educational efforts can lead to not only misunderstandings but also to potentially poor treatment compliance [75,76]. Understanding community perceptions along with the social, religious, economic and environmental context of schistosomiasis risk and risk reduction behaviors among children can inform behavior change interventions that are relevant and provide meaning to the vulnerable populations in Zanzibar [74,77,78]. A recent evaluation of a comic-strip medical booklet Juma na Kichocho associated with a schistosomiasis health education campaign in 16 primary schools in Zanzibar reported disappointing findings [44,77]. The authors recognized that changing the behaviors of children could not be done by an isolated school curriculum but needed to consider their everyday realities of daily living [77]. The information garnered from our qualitative inquiry will allow for the ideas and problem solving solutions of community members to be incorporated into a behavioral intervention that is germane to others in their communities. Increasingly, there is a commitment to bringing a community perspective into research and implementation of interventions along with a growing body of evidence that public health and health-promotion interventions based on social and behavioral science theories are more effective than those without a theoretical foundation [55,79,80]. Drawing upon the constructs of perceived seriousness, perceived susceptibility, perceived benefits, and perceived barriers from the Health Belief Model complemented by a social ecological model that addresses multiple levels of the community could provide a functional framework for designing, implementing, and evaluating a health promotion program for the prevention and control of schistosomiasis tailored to the context of community members, particularly school-aged children [17,55,74,79,81].

There were several limitations to this inquiry. Given that we used a purposive, convenience sample, the findings may not be representative of all members of the communities in which the inquiry took place, and results are not generalizable. The triangulation of data suggests that there were similarities across behaviors of school-aged children attending the government primary schools on both islands where we conducted the student discussion groups. The behaviors we assessed appeared to be generally practiced among children across the shehias on Unguja and Pemba and the lessons learned could be used to tailor messages for future schistosomiasis control programs for primary school aged children. That being said, these findings are not generalizable to children attending private schools or not attending school at all. Further investigation is needed to explore the schistosomiasis knowledge, attitudes, perceptions, and practices of students in private schools and of students who do not attend school to assess if they are similar to those from children who participated in our student discussion groups. There may have been information bias during IDI, GD, and FGD as interview subjects may have provided answers that they believed the interviewers expected or wanted to hear. Additionally, bias may have been introduced due to only having a single coder, even though data interpretations and language translations were substantiated with local research assistants.

Conducting recreational and domestic activities of daily living in water contaminated with *S. haematobium* larvae compromises the health of school-aged children in Zanzibar. An important objective of this study was to facilitate improved design of an educational and control program. Urogenital schistosomiasis, characterized as a minor illness typically of boys, along with the lack of formal school-based and community-wide education about disease transmission, symptoms, and treatment can contribute to undiagnosed disease and a lack of treatment among both girls and boys. Understanding community perceptions of disease along with the barriers and facilitators to risk reduction behaviors among children can inform

behavior change activities and health promotion programs augmented with chemotherapies for an integrated approach in support of the prevention, control, and elimination of urogenital schistosomiasis in Zanzibar and elsewhere.

Supporting Information

S1 Topic Guide. Zanzibar young students semi-structured topic guide for schistosomiasis (kichocho) drawing and discussion group. (PDF)

Acknowledgments

The authors thank all the participants who shared their knowledge and their time to educate the behavioral research team. We would also like to acknowledge the hard work and commitment of all members of the Kichocho Behavioral Teams in Unguja and Pemba who are not coauthors and our expert driver Ali "Kichocho."

Author Contributions

Conceived and designed the experiments: BP SMA KAM DR SK. Performed the experiments: BP SMA FMA JNA UAM. Analyzed the data: BP. Contributed reagents/materials/analysis tools: BP SMA KAM DR. Wrote the paper: BP SMA FMA JNA UAM KAM DR SK.

References

- 1. WHO. Schistosomiasis: number of people treated worldwide in 2013. Wkly Epidemiol Rec. 2015; 5(90): 25–32.
- Steinmann P, Keiser J, Bos R, Tanner M, Utzinger J. Schistosomiasis and water resources development: systematic review, meta-analysis, and estimates of people at risk. Lancet Infect Dis. 2006; 6(7): 411–425. PMID: <u>16790382</u>
- Gryseels B. Schistosomiasis. Infect Dis Clin North Am. 2012; 26(2): 383–397. doi: <u>10.1016/j.idc.2012.</u> <u>03.004</u> PMID: <u>22632645</u>
- Barsoum RS. Urinary schistosomiasis: Review. J Adv Res. 2013; 4: 453–459. doi: <u>10.1016/j.jare.2012</u>. <u>08.004</u> PMID: <u>25685452</u>
- Ross AG, Bartley PB, Sleigh AC, Olds GR, Li Y, Williams GM, et al. Schistosomiasis. N Engl J Med. 2002; 346(16): 1212–1220. PMID: <u>11961151</u>
- Kjetland EF, Ndhlovu PD, Gomo E, Mduluza T, Midzi N, Gwanzura L, et al. Association between genital schistosomiasis and HIV in rural Zimbabwean women. AIDS. 2006; 20(4): 593–600. PMID: <u>16470124</u>
- Colley DG, Bustinduy AL, Secor WE, King CH. Human schistosomiasis. Lancet. 2014; 383(9936): 2253–2264. doi: <u>10.1016/S0140-6736(13)61949-2</u> PMID: <u>24698483</u>
- Kjetland EF, Hegertun IE, Baay MF, Onsrud M, Ndhlovu PD, Taylor M. Genital schistosomiasis and its unacknowledged role on HIV transmission in the STD intervention studies. Int J STD AIDS. 2014; 25 (10): 705–715. doi: <u>10.1177/0956462414523743</u> PMID: <u>24621458</u>
- Stecher CW, Kallestrup P, Kjetland EF, Vennervald B, Petersen E. Considering treatment of male genital schistosomiasis as a tool for future HIV prevention: a systematic review. Int J Public Health. 2015; 60(7): 839–848. doi: <u>10.1007/s00038-015-0714-7</u> PMID: <u>26298443</u>
- WHO. Schistosomiasis: number of people treated worldwide in 2014. Wkly Epidemiol Rec. 2016; 5(91): 53–60.
- Utzinger J, Raso G, Brooker S, de Savigny D, Tanner M, Ørnbjerg N, et al. Schistosomiasis and neglected tropical diseases: towards integrated and sustainable control and a word of caution. Parasitology. 2009; 136(13): 1859–1874. doi: <u>10.1017/S0031182009991600</u> PMID: <u>19906318</u>
- Rollinson D, Knopp S, Levitz S, Stothard JR, Tchuem Tchuenté LA, Garba A, et al. Time to set the agenda for schistosomiasis elimination. Acta Trop. 2013; 128(2): 423–440. doi: <u>10.1016/j.actatropica.</u> <u>2012.04.013</u> PMID: <u>22580511</u>
- 13. WHO. Schistosomiasis: progress report 2001–2011 and strategic plan 2012–2020. Geneva: World Health Organization; 2013.

- Grimes JET, Croll D, Harrison WE, Utzinger J, Freeman MC, Templeton MR. The relationship between water, sanitation and schistosomiasis: a systematic review and meta-analysis. PLoS Negl Trop Dis. 2014; 8(12): e3296. doi: <u>10.1371/journal.pntd.0003296</u> PMID: <u>25474705</u>
- Manderson L, Aagaard-Hansen J, Allotey P, Gyapong M, Sommerfeld J. Social research on neglected diseases of poverty: continuing and emerging themes. PLoS Negl Trop Dis. 2009; 3(2): e332. doi: <u>10.</u> <u>1371/journal.pntd.0000332</u> PMID: <u>19238216</u>
- Israel BA, Schulz AJ, Parker EA, Becker AB. Review of community-based research: assessing partnership approaches to improve public health. Annu Rev Public Health. 1998; 19: 173–202. PMID: 9611617
- Nutbeam D, Harris E, Wise W. Theory in a nutshell: a practical guide to health promotion theories. 3rd ed. Sydney: McGraw-Hill; 2010.
- Viswanathan M, Ammerman A, Eng E, Gartlehner G, Lohr KN, Griffith D, et al. Community-based participatory research: assessing the evidence Rockville: Agency for Healthcare Research and Quality; 2004.
- Knopp S, Mohammed KA, Ali SM, Khamis IS, Ame SM, Albonico M, et al. Study and implementation of urogenital schistosomiasis elimination in Zanzibar (Unguja and Pemba islands) using an integrated multidisciplinary approach. BMC Public Health. 2012; 12(1): 930.
- 20. Knopp S, Person B, Ame SM, Mohammed KA, Ali SM, Khamis IS, et al. Elimination of schistosomiasis transmission in Zanzibar: baseline findings before the onset of a randomized intervention trial. PLoS Negl Trop Dis. 2013; 7: e2474. doi: 10.1371/journal.pntd.0002474 PMID: 24147165
- 21. Savioli L, Fenwick A, Rollinson D, Albonico M, Ame SM. An achievable goal: control and elimination of schistosomiasis. Lancet. 2015; 386(9995): 739.
- Barry M. Ethical considerations of human investigation in developing countries: the AIDS dilemma. N Engl J Med. 1988; 319(16): 1083–1086. PMID: <u>3173436</u>
- 23. Bhutta ZA. Beyond informed consent. Bull World Health Organ. 2004; 82(10): 771–777. PMID: 15643799
- 24. OCGS. 2012 Population and housing survey of the United Republic of Tanzania. Population distribution by administrative areas. Dar es Salaam and Zanzibar Town: National Bureau of Statistics and Office of Chief Government Statistician; 2013.
- 25. NBS. Basic demographic and socio-economic profile, Tanzania Zanzibar. Dar es Salaam: National Bureau of Statistics, Ministry of Finance; 2014.
- Savioli L, Dixon H, Kisumku UM, Mott KE. Control of morbidity due to Schistosoma haematobium on Pemba island; selective population chemotherapy of schoolchildren with haematuria to identify highrisk localities. Trans R Soc Trop Med Hyg. 1989; 83(6): 805–810. PMID: <u>2515637</u>
- Mgeni AF, Kisumku UM, McCullough FS, Dixon H, Yoon SS, Mott KE. Metrifonate in the control of urinary schistosomiasis in Zanzibar. Bull World Health Organ. 1990; 68(6): 721–730. PMID: <u>2127381</u>
- Savioli L, Hatz C, Dixon H, Kisumku UM, Mott KE. Control of morbidity due to Schistosoma haematobium on Pemba Island: egg excretion and hematuria as indicators of infection. Am J Trop Med Hyg. 1990; 43(3): 289–295. PMID: 2121056
- Stothard JR, French MD, Khamis IS, Basáñez MG, Rollinson D. The epidemiology and control of urinary schistosomiasis and soil-transmitted helminthiasis in schoolchildren on Unguja island, Zanzibar. Trans R Soc Trop Med Hyg. 2009; 103(10): 1031–1044. doi: <u>10.1016/j.trstmh.2009.03.024</u> PMID: <u>19409588</u>
- Guidi A, Andolina C, Ame SM, Albonico M, Cioli D, Juma HH. Praziquantel efficacy and long-term appraisal of schistosomiasis control in Pemba island. Trop Med Int Health. 2010; 15(5): 614–618. doi: 10.1111/j.1365-3156.2010.02488.x PMID: 20214757
- Knopp S, Stothard JR, Rollinson D, Mohammed KA, Khamis IS, Marti H, et al. From morbidity control to transmission control: time to change tactics against helminths on Unguja Island, Zanzibar. Acta Trop. 2013; 128(2): 412–422. doi: 10.1016/j.actatropica.2011.04.010 PMID: 21586268
- Colucci E. "Focus groups can be fun": the use of activity-oriented questions in focus group discussions. Qual Health Res. 2007; 17(10): 1422–1433. PMID: <u>18000081</u>
- Stothard JR, Mgeni AF, Khamis S, Seto E, Ramsan M, Rollinson D. Urinary schistosomiasis in schoolchildren on Zanzibar island (Unguja), Tanzania: a parasitological survey supplemented with questionnaires. Trans R Soc Trop Med Hyg. 2002; 96(5): 507–514. PMID: <u>12474477</u>
- Knopp S, Mohammed KA, Rollinson D, Stothard JR, Khamis IS, Utzinger J, et al. Changing patterns of soil-transmitted helminth infections in Zanzibar in the context of national control programs. Am J Trop Med Hyg. 2009; 81(6): 1071–1078. doi: <u>10.4269/ajtmh.2009.09-0377</u> PMID: <u>19996439</u>
- **35.** Glaser BG, Strauss AL. The discovery of grounded theory: strategies for qualitative research. Chicago: Aldine Publishing Company; 1967.

- **36.** Patton MQ. Qualitative Research and Evaluation Methods. Thousand Oaks: SAGE Publications; 2002.
- Kvale S, Brinkmann S. InterViews: learning the craft of qualitative research interviewing. Thousand Oaks: SAGE Publications; 2009.
- 38. Miles MB, Huberman AM. Qualitative analysis. Thousand Oaks: SAGE Publications; 1994.
- **39.** Charmaz K. Constructing grounded theory: a practical guide through qualitative analysis. Thousand Oaks: SAGE Publications; 2006.
- 40. Saldaña J. The coding manual for qualitative researchers. Thousand Oaks: SAGE Publications; 2011.
- Ryan GW, Bernard HR. Data management and analysis methods. In: Denzin NK, Lincoln YS, editors. Handbook of qualitative research. Thousand Oaks: SAGE Publications; 2000. pp. 769–803.
- 42. Morse JM. The significance of saturation. Qual Health Res. 1995; 5(2): 147-149.
- **43.** Devers KJ. How will we know "good" qualitative research when we see it? Beginning the dialogue in health services research. Health Serv Res. 1999; 34(5 Pt 2): 1153–1188.
- Stothard JR, Mook P, Mgeni AF, Khamis IS, Khamis AN, Rollinson D. Control of urinary schistosomiasis on Zanzibar (Unguja island): a pilot evaluation of the educational impact of the *Juma na Kichocho* health booklet within primary schools. Mem Inst Oswaldo Cruz. 2006; 101 (Suppl. 1): 119–124. PMID: 17308758
- 45. Mwakitalu ME, Malecela MN, Mosha FW, Simonsen PE. Urban schistosomiasis and soil transmitted helminthiases in young school children in Dar es Salaam and Tanga, Tanzania, after a decade of anthelminthic intervention. Acta Trop. 2014; 133: 35–41. doi: <u>10.1016/j.actatropica.2014.01.012</u> PMID: <u>24495630</u>
- 46. Perera M, Whitehead M, Molyneux D, Weerasooriya M, Gunatilleke G. Neglected patients with a neglected disease? A qualitative study of lymphatic filariasis. PLoS Negl Trop Dis. 2007; 1(2): e128. PMID: <u>18060080</u>
- **47.** Person B, Addiss D, Bartholomew LK, Meijer C, Pou V, Gonzálvez G, et al. A qualitative study of the psychosocial and health consequences associated with lymphedema among women in the Dominican Republic. Acta Trop. 2007; 103(2): 90–97. PMID: <u>17638614</u>
- Rosecrans K, Cruz-Martin G, King A, Dumonteil E. Opportunities for improved chagas disease vector control based on knowledge, attitudes and practices of communities in the yucatan peninsula, Mexico. PLoS Negl Trop Dis. 2014; 8(3): e2763. doi: <u>10.1371/journal.pntd.0002763</u> PMID: <u>24676038</u>
- Savioli L, Dixon H, Kisumku UM, Mott KE. Control of morbidity due to Schistosoma haematobium on Pemba island: programme organization and management. Trop Med Parasitol. 1989; 40(2): 189–194. PMID: <u>2505381</u>
- Rudge JW, Stothard JR, Basáñez MG, Mgeni AF, Khamis IS, Khamis AN, et al. Micro-epidemiology of urinary schistosomiasis in Zanzibar: local risk factors associated with distribution of infections among schoolchildren and relevance for control. Acta Trop. 2008; 105(1): 45–54. PMID: 17996207
- Midzi N, Mtapuri-Zinyowera S, Mapingure MP, Paul NH, Sangweme D, Hlerema G, et al. Knowledge attitudes and practices of grade three primary schoolchildren in relation to schistosomiasis, soil transmitted helminthiasis and malaria in Zimbabwe. BMC Infect Dis. 2011; 11: 169. doi: <u>10.1186/1471-2334-11-169</u> PMID: <u>21668948</u>
- Odiere MR, Rawago FO, Ombok M, Secor WE, Karanja DM, Mwinzi PN, et al. High prevalence of schistosomiasis in Mbita and its adjacent islands of Lake Victoria, western Kenya. Parasit Vectors. 2012; 5: 278. doi: 10.1186/1756-3305-5-278 PMID: 23206383
- Nwaorgu OC, Okeibunor J, Madu E, Amazigo U, Onyegegbu N, Evans D. A school-based schistosomiasis and intestinal helminthiasis control programme in Nigeria: acceptability to community members. Trop Med Int Health. 1998; 3(10): 842–849. PMID: <u>9809919</u>
- Rosenstock IM. The health belief model and preventive health behavior. Health Educ Behav. 1974; 2 (4): 354–386.
- 55. Painter JE, Borba CP, Hynes M, Mays D, Glanz K. The use of theory in health behavior research from 2000 to 2005: a systematic review. Ann Behav Med. 2008; 35(3): 358–362. doi: <u>10.1007/s12160-008-9042-y</u> PMID: <u>18633685</u>
- Strecher VJ, Rosenstock IM. The health belief model. In: Baum A, Newman S, Weinman J, West R, McManus C, editors. Cambridge Handbook of psychology, health and medicine. Cambridge: Cambridge University Press; 1997. pp. 113–116.
- NICE. NICE public health guidance 6: behaviour change at population, community and individual levels. London: National Institute for Health and Clinical Excellence; 2007.

- Davies P, Walker AE, Grimshaw JM. A systematic review of the use of theory in the design of guideline dissemination and implementation strategies and interpretation of the results of rigorous evaluations. Implement Sci. 2010; 5: 14. doi: <u>10.1186/1748-5908-5-14</u> PMID: <u>20181130</u>
- Michie S, Prestwich A. Are interventions theory-based? Development of a theory coding scheme. Health Psychol. 2010; 29(1): 1–8. doi: <u>10.1037/a0016939</u> PMID: <u>20063930</u>
- Michie S, Abraham C, Eccles MP, Francis JJ, Hardeman W, Johnston M. Strengthening evaluation and implementation by specifying components of behaviour change interventions: a study protocol. Implement Sci. 2011; 6: 10. doi: <u>10.1186/1748-5908-6-10</u> PMID: <u>21299860</u>
- House of Lords. Science and Technology Select Committee. 2nd report of session 2010–12. Behaviour change. London: House of Lords; 2011.
- Mwanga JR, Magnussen P, Mugashe CL, Gabone RM, Aagaard-Hansen J. Schistosomiasis-related perceptions, attitudes and treatment-seeking practices in Magu district, Tanzania: public health implications. J Biosoc Sci. 2004; 36(1): 63–81. PMID: <u>14989532</u>
- 63. Cronin T, Sheppard J, de Wildt G. Health-seeking behaviour for schistosomiasis: a systematic review of qualitative and quantitative literature. Pan Afr Med J. 2013; 16: 130. doi: <u>10.11604/pamj.2013.16</u>. <u>130.3078</u> PMID: <u>24839538</u>
- Lansdown R, Ledward A, Hall A, Issae W, Yona E, Matulu J, et al. Schistosomiasis, helminth infection and health education in Tanzania: achieving behaviour change in primary schools. Health Educ Res. 2002; 17(4): 425–433. PMID: <u>12197588</u>
- Onyango-Ouma W, Aagaard-Hansen J, Jensen BB. The potential of schoolchildren as health change agents in rural western Kenya. Soc Sci Med. 2005; 61(8): 1711–1722. PMID: <u>15967559</u>
- Mwanga JR, Jensen BB, Magnussen P, Aagaard-Hansen J. School children as health change agents in Magu, Tanzania: a feasibility study. Health Promot Int. 2008; 23(1): 16–23. PMID: <u>18086688</u>
- **67.** Milburn K. A critical review of peer education with young people with special reference to sexual health. Health Educ Res. 1995; 10(4): 407–420. PMID: <u>10159674</u>
- Prince F. The relative effectiveness of a peer-led and adult-led smoking intervention program. Adolescence. 1995; 30(117): 187–194. PMID: <u>7625255</u>
- Albrecht S, Payne L, Stone CA, Reynolds MD. A preliminary study of the use of peer support in smoking cessation programs for pregnant adolescents. J Am Acad Nurse Pract. 1998; 10(3): 119–125. PMID: 9644411
- Mellanby AR, Rees JB, Tripp JH. Peer-led and adult-led school health education: a critical review of available comparative research. Health Educ Res. 2000; 15(5): 533–545. PMID: <u>11184213</u>
- Medley A, Kennedy C, O'Reilly K, Sweat M. Effectiveness of peer education interventions for HIV prevention in developing countries: a systematic review and meta-analysis. AIDS Educ Prev. 2009; 21(3): 181–206. doi: 10.1521/aeap.2009.21.3.181 PMID: 19519235
- Parker M, Allen T, Hastings J. Resisting control of neglected tropical diseases: dilemmas in the mass treatment of schistosomiasis and soil-transmitted helminths in north-west Uganda. J Biosoc Sci. 2008; 40(2): 161–181. PMID: <u>17761005</u>
- 73. Aagaard-Hansen J, Mwanga JR, Bruun B. Social science perspectives on schistosomiasis control in Africa: past trends and future directions. Parasitology. 2009; 136(13): 1747–1758. doi: <u>10.1017/</u> <u>S0031182009006404</u> PMID: <u>19523256</u>
- Allotey P, Reidpath DD, Pokhrel S. Social sciences research in neglected tropical diseases 1: the ongoing neglect in the neglected tropical diseases. Health Res Policy Syst. 2010; 8: 32. doi: <u>10.1186/1478-4505-8-32</u> PMID: <u>20961461</u>
- 75. Parker M, Allen T. Does mass drug administration for the integrated treatment of neglected tropical diseases really work? Assessing evidence for the control of schistosomiasis and soil-transmitted helminths in Uganda. Health Res Policy Syst. 2011; 9: 3. doi: <u>10.1186/1478-4505-9-3</u> PMID: <u>21211001</u>
- 76. Knopp S, Person B, Ame SM, Ali SM, Muhsin J, Juma S, et al. Praziquantel coverage in schools and communities targeted for the elimination of urogenital schistosomiasis in Zanzibar: a cross-sectional survey. Parasit Vectors. 2016; 9(1): 5.
- Stothard IR, Khamis AN, Khamis IS, Neo CHE, Wei I, Rollinson D. Health education and the control of urogenital schistosomiasis: assessing the impact of the Juma na Kichocho comic-strip medical booklet in Zanzibar. J Biosoc Sci. 2016; doi: 10.1017/S0021932016000122
- Person B, Knopp S, Ali SM, A'Kadir FM, Khamis AN, Ali JN, et al. Community co-designed schistosomiasis control interventions for school-aged children in Zanzibar. J Biosoc Sci. 2016; doi: <u>10.1017/</u> S0021932016000067
- 79. Glanz K, Bishop DB. The role of behavioral science theory in development and implementation of public health interventions. Annu Rev Public Health. 2010; 31: 399–418. doi: <u>10.1146/annurev.publhealth.</u> <u>012809.103604</u> PMID: <u>20070207</u>

- Certain E, Terry RF, Zicker F. Shaping the research agenda. PLoS Negl Trop Dis. 2015; 9(1): e3350. doi: <u>10.1371/journal.pntd.0003350</u> PMID: <u>25569230</u>
- Stokols D. Translating social ecological theory into guidelines for community health promotion. Am J Health Promot. 1996; 10(4): 282–298. PMID: <u>10159709</u>